Author’s response to reviews

Title: Antibacterial resistance in ophthalmic infections: a multi-centre analysis across UK care settings

Authors:

Alice Lee (alice.lee13@imperial.ac.uk)
Kanchana Niruttan (kanchana.niruttan12@imperial.ac.uk)
Timothy Rawson (timothy.rawson07@imperial.ac.uk)
Luke Moore (l.moore@imperial.ac.uk)

Version: 2 Date: 24 Aug 2019

Author’s response to reviews:

Dear BMC INfectious Diseases Editorial Team,

Thank you for your further editorial comments on our manuscript. We have addressed these as below. We have attached the re-revised manuscript without tracked changes/highlighted text as requested.

With thanks

Luke

1. Please include three to ten keywords representing the main content of the article, below your abstract and above the main text of your manuscript.

DONE

2. Please change the section heading “Introduction” to “Background”.

DONE

3. As some abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.
4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.